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DAQ Status and Uptime

Far Detector
98.8%  POT-weighted 
uptime for last 4 weeks

Near Detector
99.6% POT-weighted 
uptime for last 4 weeks
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Highlights

● Far Detector:
– During beam downtime for septa repair (4/4), performed 

hardware swaps recover a few FEBs
● Near Detector:

– ~9:30 this morning, MINOS chiller water temperature alarms 
went of

● Chiller was taken of line to clean filters and backup chiller is not 
functioning at full capacity

– We were not aware of this regular maintenance – request notification in the future 

– NOvA is very sensitive to changes in water temperature
● Near Detector run was stopped and HV turned of
● Chiller is back on line, but recovery is still ongoing

– When HV is lost, detector can take up to a day for noise to reach acceptable levels
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Computing
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